\documentclass[12pt]{article}
%\setlength{\oddsidemargin}{0.25in}
%\setlength{\evensidemargin}{0.25in}
%\setlength{\textwidth}{6in}
%\setlength{\headsep}{0in}
%\setlength{\headheight}{0in}
%\setlength{\topmargin}{0.25in}
%\setlength{\textheight}{8.5in}
\usepackage{e-jc}

\usepackage{latexsym, amsthm, amsmath, enumerate}
\usepackage{graphicx, xspace, ifthen, rotating, pict2e}
\usepackage[svgnames]{xcolor}
\usepackage[charter]{mathdesign}
\usepackage{algpseudocode, algorithmicx}

\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

\usepackage{young}
\ysetshade{Blue!30}
\ysetaltshade{Green!50}
\YSetShade{Green!30}
\YSetAltShade{Purple!45}
%\YSETSHADE{Purple!31}
%\YSETALTSHADE{Orange!37}


\title{Minimal orbits of promotion}
\author{Kevin Purbhoo\thanks{%
Research of Purbhoo was 
supported by an NSERC Discovery grant.}\\
\small Combinatorics and Optimization Department \\[-.8ex]
\small University of Waterloo \\[-.8ex]
%\small 200 University Ave. W. \\[-.8ex]
\small Waterloo ON, Canada \\
\small \texttt{kpurbhoo@uwaterloo.ca}\\
\and 
Donguk Rhee \\
\small Lunarch Studios \\[-.8ex]
\small 144 Park Street, Unit 1702 \\[-.8ex]
\small Waterloo ON, Canada. \\
\small \texttt{david@lunarchstudios.com}.
} 

\date{\dateline{Jan 22, 2016}{XX} \\
\small Mathematics Subject Classifications: 05E10, 05A19}

\setcounter{MaxMatrixCols}{20}
\allowdisplaybreaks[1]

\newcommand{\bigmid}{\ \big|\ }
\newcommand{\Bigmid}{\ \Big|\ }
\newcommand{\biggmid}{\ \bigg|\ }
\newcommand{\Biggmid}{\ \Bigg|\ }
\newcommand{\middlemid}{\ \middle|\ }

\newcommand{\frn}{\mathfrak{n}}
\newcommand{\frb}{\mathfrak{b}}
\newcommand{\frg}{\mathfrak{g}}
\newcommand{\frk}{\mathfrak{k}}
\newcommand{\frt}{\mathfrak{t}}
\newcommand{\CC}{\mathbb{C}}
\newcommand{\FF}{\mathbb{F}}
\newcommand{\ZZ}{\mathbb{Z}}
\newcommand{\QQ}{\mathbb{Q}}
\newcommand{\RR}{\mathbb{R}}
\newcommand{\FP}{\mathbb{FP}}
\newcommand{\RP}{\mathbb{RP}}
\newcommand{\PP}{\mathbb{P}}
\newcommand{\VV}{\mathbb{V}}
\newcommand{\KK}{\mathbb{K}}
\newcommand{\KP}{\mathbb{KP}}


\newcommand{\Hy}{\widehat y}
\newcommand{\Hw}{{\widehat w}}
\newcommand{\HU}{{\widehat U}}
\newcommand{\Hsigma}{{\widehat \sigma}}
\newcommand{\la}[1]{{\lambda_{#1}}}
\newcommand{\veela}[1]{{\smash{\lambda^\vee}\!\!\!_{#1}}}
\newcommand{\primela}[1]{{\smash{\lambda'}\!_{#1}}}
\newcommand{\Tlambda}{{\widetilde \lambda}}
\newcommand{\Tkappa}{{\widetilde \kappa}}
\newcommand{\Tmu}{{\widetilde \mu}}
\newcommand{\Tnu}{{\widetilde \nu}}
\newcommand{\Tsigma}{{\widetilde \sigma}}
\newcommand{\Tx}{\widetilde x}
\newcommand{\Tw}{\widetilde w}
\newcommand{\Tboldw}{\widetilde \boldw}
\newcommand{\TF}{\widetilde F}
\newcommand{\redX}{\overline{X}}
\newcommand{\redF}{\overline{F}}
\newcommand{\calA}{\mathcal{A}}
\newcommand{\redA}{\overline{\calA}}
\newcommand{\redL}{\overline{\Lambda}}
\newcommand{\redcell}{\redX\vphantom{X}^\circ}

\newcommand{\nth}{\ensuremath{^\text{th}}\xspace}

\newcommand{\puiseux}[1]{\CC\{\!\{#1\}\!\}}
\newcommand{\Fpuiseux}[1]{\FF\{\!\{#1\}\!\}}
\newcommand{\mirrc}{\mathcal{R}}
\newcommand{\calU}{\mathcal{U}}
\newcommand{\orbit}{\mathcal{O}}
\newcommand{\Orbits}{\mathrm{Orb}}

\newcommand{\poln}{\CC_{n-1}[z]}
\newcommand{\Rpoln}{\RR_{n-1}[z]}
\newcommand{\Kpoln}{\KK_{n-1}[z]}
\newcommand{\Fpoln}{\FF_{n-1}[z]}
\newcommand{\pol}[1]{\CC_{#1}[z]}
\newcommand{\Rpol}[1]{\RR_{#1}[z]}
\newcommand{\Kpol}[1]{\KK_{#1}[z]}
\newcommand{\Fpol}[1]{\FF_{#1}[z]}
\newcommand{\Ppol}{\PP\big(\pol{2M}\big)}
\newcommand{\KPpolN}{\PP(\KK_N[z])}


\newcommand{\SL}{\mathrm{SL}}
\newcommand{\Sp}{\mathrm{Sp}}
\newcommand{\Ogroup}{\mathrm{O}}
\newcommand{\PGL}{\mathrm{PGL}}
\newcommand{\GL}{\mathrm{GL}}
\newcommand{\Wr}{\mathrm{Wr}}
\newcommand{\Gr}{\mathrm{Gr}}
\newcommand{\LG}{\mathrm{LG}(n)}
\newcommand{\OG}{\mathrm{OG}(n,2n{+}1)}
%\newcommand{\Rect}{{\mathchoice%
%{\raisebox{.05ex}{$\sqsubset\!\!\!\sqsupset$}}
%{\raisebox{.05ex}{$\sqsubset\!\!\!\sqsupset$}}
%{\sqsubset\!\!\!\sqsupset}
%{\sqsubset\!\!\!\sqsupset}
%}}
\newcommand{\smallrect}{{\mathchoice%
   {\raisebox{.05ex}{\footnotesize $\sqsubset\!\!\!\sqsupset$}}
   {\raisebox{.05ex}{\footnotesize $\sqsubset\!\!\!\sqsupset$}}
   {\raisebox{.1ex}{\tiny $\sqsubset\!\!\!\sqsupset$}}
   {\raisebox{.1ex}{\tiny $\sqsubset\!\!\!\sqsupset$}}
}}
\newcommand{\lowBox}{{\mathchoice%
{{\raisebox{-0.2ex}{$\Box$}}}
{{\raisebox{-0.2ex}{$\Box$}}}
{\Box}
{\Box}
}}
\newcommand{\shiftedstair}
{\,{\setlength\unitlength{1.5ex}%
\begin{picture}(1,1)(0,0.25)
\thicklines
\put(0,1){\line(1,-1){1}}
\put(0,1){\line(1,0){1}}
\put(1,0){\line(0,1){1}}
\end{picture}%
}\,}
\newcommand{\stair}
{\,{\setlength\unitlength{1.5ex}%
\begin{picture}(1,1)(0,0.25)
\thicklines
\put(0,0){\line(1,1){1}}
\put(0,1){\line(1,0){1}}
\put(0,0){\line(0,1){1}}
\end{picture}%
}\,}
\newcommand{\SYT}{\mathsf{SYT}}
\newcommand{\DST}{\mathsf{DST}}
\newcommand{\AST}{\mathsf{AST}}
\newcommand{\ribbon}{\mathsf{SRT}^r}
\newcommand{\ribbonplain}{\mathsf{SRT}}
\newcommand{\rotribbon}{\mathsf{RRT}^r}
\newcommand{\rotribbonplain}{\mathsf{RRT}}
\newcommand{\longribbon}{\widehat{\mathsf{RRT}}\,\!^r}
\newcommand{\longribbonplain}{\widehat{\mathsf{RRT}}\,\!}

\newcommand{\calT}{\mathcal{T}}
\newcommand{\calM}{\mathcal{M}}
\newcommand{\calO}{\mathcal{O}}
\newcommand{\bolda}{{\bf a}}
\newcommand{\boldb}{{\bf b}}
\newcommand{\boldc}{{\bf c}}
\newcommand{\boldk}{{\bf k}}
\newcommand{\bolds}{{\bf s}}
\newcommand{\boldp}{{\bf p}}
\newcommand{\boldw}{{\bf w}}
\newcommand{\boldx}{{\bf x}}
\newcommand{\boldy}{{\bf y}}
\newcommand{\boldlambda}{{\mathchoice
{\mbox{\boldmath{$\lambda$}}}
{\mbox{\boldmath{$\lambda$}}}
{\mbox{\scriptsize\boldmath{$\lambda$}}}
{\mbox{\scriptsize\boldmath{$\lambda$}}}
}}
\newcommand{\boldalpha}{{\mathchoice
{\mbox{\boldmath{$\alpha$}}}
{\mbox{\boldmath{$\alpha$}}}
{\mbox{\scriptsize\boldmath{$\alpha$}}}
{\mbox{\scriptsize\boldmath{$\alpha$}}}
}}
\newcommand{\ordT}{\mathsf{T}}
\newcommand{\derz}{{\textstyle \frac{d}{dz}}}
\newcommand{\smallidmatrix}
{\left(\begin{smallmatrix} 1 & 0 \\ 0 & 1\end{smallmatrix}\right)}
\newcommand{\stdreflect}
{\left(\begin{smallmatrix} 0 & 1 \\ 1 & 0\end{smallmatrix}\right)}
\newcommand{\altreflect}
{\left(\begin{smallmatrix} -1 & 0 \\ 0 & 1\end{smallmatrix}\right)}
\newcommand{\halfrotate}
{\left(\begin{smallmatrix} 0 & -1 \\ 1 & 0\end{smallmatrix}\right)}
\newcommand{\quarterrotate}
{\left(\begin{smallmatrix} 1 & -1 \\ 1 & 1\end{smallmatrix}\right)}
\newcommand{\imag}{{\sqrt{\!-\!1}}}

\newcommand{\mindeg}{\mathop{\mathrm{mindeg}}}
\newcommand{\Proj}{\mathop{\mathrm{Proj}}}
\newcommand{\Spec}{\mathop{\mathrm{Spec}}}
\newcommand{\codim}{\mathop{\mathrm{codim}}}
\newcommand{\rcore}{\mathrm{core}^r}
\newcommand{\rspec}{\mathrm{spec}^r}
\newcommand{\aug}{\mathrm{aug}}
\newcommand{\sgn}{\mathrm{sgn}}
\newcommand{\ord}{\mathrm{ord}}
\newcommand{\negate}{\mathrm{neg}}
\newcommand{\slide}{\mathrm{slide}}
\newcommand{\fold}{\mathsf{fold}}
\newcommand{\switch}{\mathsf{switch}}
\newcommand{\rectify}{\mathrm{rect}}
\newcommand{\mult}{\mathrm{mult}}
%\newcommand{\promote}{{\boldsymbol{j}}}
\newcommand{\promote}{\partial}
\newcommand{\trunc}{\mathrm{Trunc}}
\newcommand{\evac}{\textbf{\textit{e}}}
\newcommand{\val}{\mathrm{val}}
\newcommand{\leadterm}{\text{\rm \footnotesize LT}}
\newcommand{\leadcoeff}{\text{\rm \footnotesize LC}}
\newcommand{\initial}{\mathrm{In}}
\newcommand{\weight}{\mathrm{wt}}
\newcommand{\height}{\mathrm{ht}}
\newcommand{\identity}{\mathrm{id}}
\newcommand{\sign}{\mathrm{sign}}

\newcommand{\one}{{\mathchoice%
{\yng[bb][1ex](1)}
{\yng[bb][1ex](1)}
{\yng[bb][.7ex](1)}
{\yng[bb][.7ex](1)}
}}
\newcommand{\diag}{{\yeveryframe{\yshade}\one}}
\newcommand{\Rect}{{\setlength{\yframethickness}{.7pt}\yhwratio{2:3}%
\mathchoice%
{{\yng[bb][1.1ex](1)}}
{{\yng[bb][1.1ex](1)}}
{{\yng[bb][.7ex](1)}}
{{\yng[bb][.7ex](1)}}
}}
\newcommand{\Hone}{{\widehat{\one}}}
\newcommand{\boxseq}{{\setlength{\yframethickness}{.25ex}
{\yng[bb][.75ex](1)}\,}}
\newcommand{\SYTRect}{\SYT\big(\Rect\big)}
%\mathchoice%
%{\begin{young}[bb][1ex] ! \end{young}}
%{\begin{young}[bb][1ex] ! \end{young}}
%{\begin{young}[bb][.7ex] ! \end{young}}
%{\begin{young}[bb][.7ex] ! \end{young}}
%}}

\newcommand{\richvar}[1]{\widetilde X_{\widetilde\lambda_{#1}/\widetilde\mu_{#1}}}
\newcommand{\richid}[1]{I_{\widetilde\lambda_{#1}/\widetilde\mu_{#1}}}

\newenvironment{packedenumi}{
\begin{enumerate}[(i)]
  \setlength{\itemsep}{0pt}
  %\setlength{\parskip}{0pt}
  %\setlength{\parsep}{0pt}
}{\end{enumerate}}

\newenvironment{packedenum}{
\begin{enumerate}
  \setlength{\itemsep}{0pt}
  %\setlength{\parskip}{0pt}
  %\setlength{\parsep}{0pt}
}{\end{enumerate}}

\newenvironment{alignqed}
  {\begin{displaymath}\begin{aligned}[b]}
  {\end{aligned}\qedhere\end{displaymath}}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}[lemma]{Theorem}
\newtheorem{fact}[lemma]{Fact}
\newtheorem{proposition}[lemma]{Proposition}
\newtheorem{corollary}[lemma]{Corollary}

\newenvironment{restatetheorem}[1]
   {\begingroup \newtheorem*{theoremx}{#1}\begin{theoremx}\em}
   {\end{theoremx}\endgroup}

\theoremstyle{definition}
\newtheorem{example}[lemma]{Example}
\newtheorem{definition}[lemma]{Definition}
\newtheorem{remark}[lemma]{Remark}
\newtheorem{condition}[lemma]{Condition}
\newtheorem{convention}[lemma]{Convention}

\newtheorem{algorithm}{Algorithm}
\renewcommand*{\thealgorithm}{\Alph{algorithm}}

%\numberwithin{equation}{section}
%\numberwithin{figure}{section}
%\numberwithin{table}{section}

\definecolor{DarkBlue}{rgb}{0, 0.1, 0.55}
\definecolor{DarkRed}{rgb}{0.45, 0, 0}
%\newcommand{\defn}[1]{\textbf{\color{DarkBlue} \emph{#1}}}
\newcommand{\defn}[1]{\textbf{#1}}


\newcommand{\comment}[1]{\textbf{\color{Green} $\star$ #1 $\star$}}

\begin{document}
\begin{thebibliography}{00}

\bibitem{BSS}
G. Benkart, F. Sottile and J. Stroomer,
\emph{Tableau Switching: Algorithms and Applications},
J. Combin. Theory Ser. A, \textbf{76} (1996) no.~1, 11--43.

\bibitem{FK} B. Fontaine and J. Kamnitzer,
\emph{Cyclic sieving, rotation, and geometric representation theory},
Selecta Math. \textbf{20} (2014) no.~2, 609--624.

\bibitem{Ful} W. Fulton,
\emph{Young tableaux with applications to representation theory and geometry},
Cambridge U.P., New York, 1997.

\bibitem{Gre} C. Greene,
\emph{An extension of Schensted's theorem},
Adv. Math., \textbf{14} (1974), 254--265.

\bibitem{Hai} M. Haiman,
\emph{Dual equivalence with applications, including a conjecture
of Proctor}, Discrete Math. \textbf{99} (1992), 79--113.


\bibitem{LLT} A. Lascoux, B. Leclerc, and J.-Y. Thibon,
\emph{The plactic monoid} in ``Algebraic combinatorics on words'',
Cambridge U.P. Cambridge (2002), 164--196.

\bibitem{LS} A. Lascoux, M. P. Sch\"utzenberger,
\emph{Le monoide plaxique},
in ``Combinatorial Mathematics, Optimal Designs and Their Applications'', 
Ann. Discrete Math., \textbf{6} (1980), 251--255.

\bibitem{PPR}
T. K. Petersen, P. Pylyavskyy and B. Rhoades,
\emph{Promotion and cyclic sieving via webs},
J. Alg. Combin., \textbf{30} (2009) no. 1, 19--41.

\bibitem{Pur-ribbon}
K. Purbhoo,
\emph{Wronskians, cyclic group actions, and ribbon tableaux},
Trans. Amer. Math. Soc. \textbf{365} (2013), 1977--2030.

\bibitem{Rhe}
D. Rhee,
\emph{Cyclic Sieving Phemonon of Promotion on Rectangular Tableaux},
Master's Thesis, University of Waterloo, 2012.

\bibitem{Rho}
B. Rhoades,
\emph{Cyclic sieving, promotion, and representation theory},
J. Combin. Theory Ser. A, \textbf{117} (2010) no. 1, 38--76.

\bibitem{Sag-CSP}
B. Sagan,
\emph{The cyclic sieving phenomenon: a survey},
London Math. Soc. Lecture Note Ser. \textbf{392}
(2011), 183--234.

\bibitem{Sch}
M.-P.~Sch\"utzenberger,
\emph{La correspondance de Robinson}, in
Combinatoire et repr\'esentation du groupe sym\'etrique,
Lecture Notes in Math., \textbf{579} (1977), Springer-Verlag,
59--113.


\bibitem{SRW}
D. Stanton, V. Reiner and D. White,
\emph{The cyclic sieving phenomenon},
J. Comb. Theorey. Ser. A., \textbf{108} (2004), 17--50.
%\bibitem{SW}
%D. Stanton and D. White,
%\emph{A Schensted algorithm for rim-hook tableaux},
%J. Combin. Theory Ser. A, \textbf{40} (1985) 211--247.
%
%\bibitem{Sta}
%R. Stanley,
%\emph{Enumerative combinatorics}, vol. 2, 
%Cambridge University Press, Cambridge, 1999.

\bibitem{Sta}
R. P. Stanley,
\emph{Promotion and evacuation},
Elec. J. Combin. \textbf{16} (2009) no. 2, R9 (electronic).

\bibitem{Wes}
B. Westbury,
\emph{Invariant tensors and the cyclic sieving phenomenon},
preprint, \texttt{arXiv:\allowbreak0912.1512}.
\end{thebibliography}

\maketitle

\begin{abstract}
We give a
bijection between the symmetric group $S_n$, and the set of
standard Young tableaux of rectangular shape $m^n$, $m \geq n$,
that have order $n$ under \emph{jeu de taquin} promotion.
\end{abstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction}

Fix positive integers $m \geq n$, and let $\Rect$ be either
the $m \times n$ rectangle, or the $n \times m$ rectangle.
The \defn{promotion} map $\promote: \SYTRect \to \SYTRect$
defines an action of $(\ZZ, +)$ 
on the set of standard Young tableaux of shape $\Rect$\,.  
For $T \in \SYT(\Rect)$,
$\promote T$ is computed by deleting the entry $1$ from $T$, 
decrementing each entry by $1$, rectifying,
and finally adding an entry $mn$ in the lower-right corner.
For example,
\[
T \,=\,
\begin{young}[c]
1 & 2 & 5  \\
3 & 4 & 6 
\end{young} 
\,\to\,
\begin{young}[c]
  & 2 & 5  \\
3 & 4 & 6 
\end{young} 
\,\to\,
\begin{young}[c]
  & 1 & 4  \\
2 & 3 & 5 
\end{young} 
\,\to\,
\begin{young}[c]
1 & 3 & 4  \\
2 & 5 &   
\end{young} 
\,\to\,
\begin{young}[c]
1 & 3 & 4  \\
2 & 5 & 6 
\end{young} 
\,=\, \promote T
 \,.
\]
Interest in this action stems from its
connections to geometry and representation theory, and its
striking combinatorial properties
(see \cite{FK, Pur-ribbon, Rho, Sta, Wes}).

Let $\calO_r := \{T \in \SYTRect \mid \promote^r T = T\}$ denote
the set of tableaux whose order under promotion divides $r$.
By a theorem of Haiman~\cite{Hai},
$\promote^{mn} T = T$ for all $T \in \SYTRect$; hence
$\calO_r$ is empty if $r$ is coprime to $mn$.
It is also not hard to show that $\calO_r$ is empty for $r < n$;
this is implicit in the proof of Theorem~\ref{thm:bijection}.
The minimal orbits of promotion, therefore, have order $n$.

The action of promotion on $\SYTRect$ exhibits a 
cyclic sieving phenomenon, as defined in \cite{SRW}: 
we have $|\calO_r| = F(\zeta^r)$, 
where
$F(q)$ is a $q$-analogue of the hook 
length formula for $|\SYTRect|$, 
and $\zeta$ is a primitive $(mn)$\nth root of unity.
The quantity $F(\zeta^r)$ appears in
a number of other places in representation theory and combinatorics,
which proffers a variety of avenues of proof for this cyclic sieving
theorem.  It was first proved by Rhoades 
using Kazhdan-Lusztig 
theory \cite{Rho}. Subsequently other proofs were found using 
representation theory 
of $\SL_n$ 
\cite{Wes}, 
and the geometry of the Grassmannian \cite{Pur-ribbon} and
the affine Grassmannian~\cite{FK}.
Simpler, more combinatorial proofs are known in special cases 
when $n =2, 3$ \cite{PPR}.
The survey~\cite{Sag-CSP} discusses of a number of related results.
However, at present there is no known combinatorial proof in general, 
nor any proof that gives an \emph{effective}
description of the sets $\calO_r$.

The purpose of this paper is to give an explicit combinatorial 
construction 
of the orbits in $\calO_n$, i.e. the minimal orbits of promotion.
We will assume that the reader is familiar with basic
definitions from the combinatorial theory of Young tableau, 
such as Schensted insertion,
\emph{jeu de taquin} operations (sliding, reverse-sliding, rectification),
Knuth equivalence and dual equivalence; all of these may be 
found in \cite{Ful}.

Using Rhoades' cyclic sieving theorem, 
one can compute that $|\calO_n| = n!$.
Our main result gives a bijection between the symmetric 
group $S_n$ and $\calO_n$.  Under this bijection promotion corresponds
to right-multiplication by the $n$-cycle $(1\ n\ n{-}1\ \dots\ 2)$.
%The proof is purely combinatorial, and does not use Rhoades'
%cyclic sieving theorem; hence this gives an independent proof of the 
%fact that $|\calO_n| = n!$.
There are a number of arbitrary choices involved in constructing the 
bijection, and much of the proof is concerned with showing that the 
construction is in fact well-defined.

To begin, choose a skew shape $\la+/\la- \subset \Rect$\,, consisting
of $n$ boxes $\diag_1, \diag_2, \dots, \diag_n$,  such that 
$\diag_{i+1}$ is strictly above and strictly right 
of $\diag_i$, for $i = 1, \dots, n-1$.
We call $\la+/\la-$ a \defn{diagonal} of $\Rect$\,,
(see Figure~\ref{fig:diagonal}).
For each permutation $w \in S_n$, we define a tableau $T_w^\la+$ 
of shape $\la+$, using a procedure similar to rectification.  
In the following algorithm, $T$ is a tableau
under construction.  If $\one$ is a box of $\la+$, we
write $\one \in \la+$, and $T[\one]$ 
denotes the entry of $T$ in box $\one$.

\begin{figure}[tb]
\begin{center}
\begin{young} 
 & & & &!& \\
 & & &!& & \\
 & &!& & & \\
!& & & & & \\
\end{young}
\end{center}
\caption{An example of a diagonal of $\protect\Rect$\,: here 
$n = 4$, $m=6$, $\la+ = 5431$ and $\la- = 432$.}
\label{fig:diagonal}
\end{figure}

\begin{algorithm}
\label{alg:forward} \emph{INPUT:} A permutation $w \in S_n$.
\begin{algorithmic}
\State
Begin with
$T[\diag_i] := w(i)$, for $i = 1, \dots, n$, leaving
all boxes of $\la-$ unfilled; 
\While{$\mathrm{shape}(T) \neq \la+$}
\State  Let $\mu \subset \la+$ be the unfilled boxes to the left of $T$; 
\State  Choose any corner box $\one \in \mu$; 
\State  Let $T'$ be the tableau obtained by sliding $\one$ through $T$; 
\State  If the final position of the sliding path is $\diag_i$, then set 
$T'[\diag_i] := T[\diag_i] + n$; 
\State  Set $T := T'$;
\EndWhile
\State \Return the resulting tableau, $T_w^\la+ := T$.
\end{algorithmic}
\end{algorithm}

An example of Algorithm~\ref{alg:forward} 
is given in Figure~\ref{fig:forward}.
Note that in this example, $T_w^\la+$ is not a standard Young tableau, 
since the entries are not $\{1, \dots, |\la+|\}$.


\begin{figure}[tb]
\begin{gather*}
{\begin{young}[c]
   &   &   &   &!2 \\
   &   &   &!4  \\
   &   &!1 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
   &   &   &   &!2 \\
   &   & 1 &!4  \\
   &   &!5 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
   &   &   &   &!2 \\
   &   & 1 &!4  \\
   & 5 &!9 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
   &   &   & 2 &!6 \\
   &   & 1 &!4  \\
   & 5 &!9 \\
!3
\end{young}} \\[1.5ex]
\ \to \ %
{\begin{young}[c]
   &   & 1 & 2 &!6 \\
   &   & 4 &!8  \\
   & 5 &!9 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
   &   & 1 & 2 &!6 \\
   &   & 4 &!8  \\
 3 & 5 &!9 \\
!7
\end{young}}
\ \to \ %
{\begin{young}[c]
   &   & 1 & 2 &!6 \\
   & 4 & 8 &!12  \\
 3 & 5 &!9 \\
!7
\end{young}} \\[1.5ex]
\ \to \ %
{\begin{young}[c]
   & 1 & 2 & 6 &!10 \\
   & 4 & 8 &!12  \\
 3 & 5 &!9 \\
!7
\end{young}}
\ \to \ %
{\begin{young}[c]
   & 1 & 2 & 6 &!10 \\
 3 & 4 & 8 &!12  \\
 5 & 9 &!13 \\
!7
\end{young}}
\ \to \ %
{\begin{young}[c]
 1 & 2 & 6 & 10 &!14 \\
 3 & 4 & 8 &!12  \\
 5 & 9 &!13 \\
!7
\end{young}}
\end{gather*}
\caption{Construction of $T_w^\la+$, with $w = 3142$, and 
$\la+ = 5431$.}
\label{fig:forward}
\end{figure}

The key difference is between 
Algorithm~\ref{alg:forward} and rectification is
that as we slide empty boxes of $\la-$ through $T$, we are also 
refilling the boxes of $\la+/\la-$.  
If we were not refilling these boxes, we would be performing 
ordinary rectification on a tableau with reading word $w$, 
which produces the insertion tableau of $w$.
Since this difference involves only entries greater that $n$, 
the insertion tableau of $w$ is the subtableau  $T_w^\la+$ formed
by entries $1, \dots, n$.
%Hence Algorithm~\ref{alg:forward} can
%be regarded as an extension of the Schensted insertion algorithm.

\begin{theorem}
\label{thm:welldefined}
The definition of $T_w^\la+$ is independent of the choices in 
Algorithm~\ref{alg:forward}.
\end{theorem}

Theorem~\ref{thm:welldefined} is analogous to the well known fact 
that ordinary rectification is well-defined \cite{Sch}.
However, despite the apparent similarity between 
Algorithm~\ref{alg:forward}
and rectification, one cannot easily deduce one fact from the other.
We discuss some of the difficulties in Section~\ref{sec:remarks}.

Similarly, we define a tableau $T_w^{\Rect/\la-}$ of
shape $\Rect/\la-$, using reverse slides.

\begin{algorithm}
\label{alg:reverse} \emph{INPUT:} A permutation $w \in S_n$.
\begin{algorithmic}
\State
Begin with $T[\diag_i] := w(i) + (m-1)n$, for $i = 1, \dots, n$, 
and all boxes of $\Rect/\la+$ unfilled; 
\While{$\mathrm{shape}(T) \neq \Rect/\la-$}
\State  Let $\mu \subset \Rect/\la-$ be the unfilled boxes to the 
  right of $T$; 
\State  Choose any corner box $\one$ of $\mu$; 
\State  Let $T'$ be the tableau obtained by reverse-sliding $\one$ through $T$; 
\State  If final position of the sliding path is $\diag_i$, then set 
$T'[\diag_i] := T[\diag_i] - n$; 
\State  Set $T := T'$; 
\EndWhile
\State \Return the resulting tableau, $T_w^{\Rect/\la-} := T$.
\end{algorithmic}
\end{algorithm}

\begin{figure}[tb]
\begin{gather*}
{\begin{young}[c]
 ,  & , & ,  & ,  &!22 & \\
 ,  & , & ,  &!24 &    & \\
 ,  & , &!21 &    &    & \\
!23 &   &    &    &    &
\end{young}}
\ \to \ %
{\begin{young}[c]
 ,  & , & ,  & ,  &!22 & \\
 ,  & , & ,  &!20 &    & \\
 ,  & , &!21 & 24 &    & \\
!23 &   &    &    &    &
\end{young}}
\ \to \ %
{\begin{young}[c]
 ,  & , & ,  & ,  &!22 & \\
 ,  & , & ,  &!20 &    & \\
 ,  & , &!21 & 24 &    & \\
!19 & 23 &    &    &    &
\end{young}} \\[1.5ex]
\ \to \ %
{\begin{young}[c]
 ,  & , & ,  & ,  &!22 & \\
 ,  & , & ,  &!20 &    & \\
 ,  & , &!21 & 24 &    & \\
!15 & 19 & 23 &    &    &
\end{young}}
\ \to \ \  \dots \ \ \to \ %
{\begin{young}[c]
 ,  & , & ,  & ,  & !14  &18  \\
 ,  & , & ,  &!12 & 16 & 20  \\
 ,  & , &!13 & 17 & 21 & 22  \\
!7  &11 & 15 & 19 & 23 & 24
\end{young}}
\end{gather*}
\caption{Construction of $T_w^{\protect\Rect/\la-}$, with 
$w = 3142$, $m = 6$, and $\la+ = 5431$.}
\label{fig:reverse}
\end{figure}

An example is given in Figure~\ref{fig:reverse}.
Since Algorithm~\ref{alg:reverse} is essentially 
``Algorithm~\ref{alg:forward} turned 
upside-down'', Theorem~\ref{thm:welldefined} implies that 
the definition of $T_w^{\Rect/\la-}$ is independent of choices.

We combine these two constructions to
produce a tableau $T_w$ of shape $\Rect$\,:
for each box $\one \in \Rect$\,, let
\[
T_w[\one] := 
\begin{cases}
T_w^\la+[\one] &\quad \text{if }\one \in \la+ \\
T_w^{\Rect/\la-}[\one] &\quad \text{otherwise.}
\end{cases}
\]
For example, for $w=3142$, $m=6$, we combine the tableaux in 
Figures~\ref{fig:forward} and~\ref{fig:reverse} to 
obtain
\[
    T_w = 
{\begin{young}[c]
 1  & 2 & 6  & 10  & 14  &18  \\
 3  & 4 & 8  & 12 & 16 & 20  \\
 5  & 9 & 13 & 17 & 21 & 22  \\
 7  &11 & 15 & 19 & 23 & 24
\end{young}}\ .
\]

Since the definition of $T_w$ is piecewise, it is not immediately
clear that this is always a sensible construction.  We will show
that the constructions in Algorithms~\ref{alg:forward} 
and~\ref{alg:reverse} agree on the diagonal, i.e. 
$T_w^\la+[\diag_i] = T_w^{\Rect/\la-}[\diag_i]$,
for $i=1, \dots, n$.  This is the first step in proving:

\begin{theorem}
\label{thm:combine}
$T_w$ is a standard Young tableau.  Moreover, the definition of $T_w$ is 
independent of the choice of diagonal $\la+/\la-$.
\end{theorem}

Our main result states that this construction gives the minimal
orbits of promotion.

\begin{theorem}
\label{thm:bijection}
The map $w \mapsto T_w$ defines a bijection between $S_n$ and $\calO_n$.
Specifically the following hold:
\begin{packedenumi}
\item  For all $w \in S_n$, $\promote T_w= T_{wc}$, where 
$c = (1\ n\ n{-}1\ \dots\ 2)$.
In particular $T_w \in \calO_n$.
\item If $w, w' \in S_n$ and $w(i) \neq w'(i)$, then 
$T_w[\diag_i] \neq T_{w'}[\diag_i]$.  In particular $w \mapsto T_w$
is injective.
\item For each $T \in \calO_n$, consider the function
$w : \{1, \dots, n\} \to \{1, \dots, n\}$
such that $ w(i) \equiv T[\diag_i] \pmod n$, for $i =1, \dots, n$.  
We have $w \in S_n$, and $T_w = T$.  In particular $w \mapsto T_w$
is surjective.
\end{packedenumi}
\end{theorem}

The rest of this paper is organized as follows.  In 
Section~\ref{sec:strategy} we develop a reduction strategy for 
proving Theorems~\ref{thm:welldefined} and~\ref{thm:combine}.
This strategy is implemented in Section~\ref{sec:descents}, where we
prove two lemmas: the first reducing the problem to one we can solve, 
and the second solving it.  All three 
theorems 
%Theorems~\ref{thm:welldefined}, \ref{thm:combine} and~\ref{thm:bijection}
are proved in Section~\ref{sec:proofs}.  Finally, in 
Section~\ref{sec:remarks} 
we discuss some additional facts that are true, and some that
we would like to be true.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Strategy}
\label{sec:strategy}

To prove Theorem~\ref{thm:welldefined}, we need to formulate it
in a different way.  As with ordinary rectification, each 
possible sequence of choices of boxes in Algorithm~\ref{alg:forward} 
can be encoded by a standard Young tableau $U \in \SYT(\la-)$,
by putting $U[\one] := |\la-|+1-k$ if $\one$ is the box chosen
in the $k$\nth iteration of the loop.  Let $T_w^U$ denote
the result of applying Algorithm~\ref{alg:forward} with
order of boxes encoded by $U$.  For example, Figure~\ref{fig:forward}
computes $T_{3142}^U$, with
\[
    U \ =\ %
\begin{young}[c]
1 & 3 & 6 & 7 \\
2 & 4 & 9 \\
5 & 8
\end{young}\,.
\]
Theorem~\ref{thm:welldefined}
states that $T_w^U$ is independent of $U \in \SYT(\la-)$.


\begin{figure}[tb]
\begin{gather*}
{\begin{young}[c]
 1 & 3 & 6 & 7 &!2 \\
 2 & 4 & 9 &!4  \\
 5 & 8 &!1 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & 1 & 3 & 7 &!2 \\
 2 & 4 & 6 &!4  \\
 5 & 8 &!5 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & 1 & 3 &!6 \\
 2 & 4 & 6 &!4  \\
 5 & 8 &!5 \\
!3
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & 1 & 3 &!6 \\
?3 & 4 & 6 &!4  \\
 2 & 8 &!5 \\
!7
\end{young}}
\\[1.5ex]
\ \to \ %
{\begin{young}[c]
?1  & ?2 & 1 & 3 &!6 \\
?3 & ?4  & 4 & !8  \\
 2 & 8 &!5 \\
!7
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & 1 & 3 &!6 \\
?3 & ?4  & 4 & !8  \\
?5 & 2 & !9 \\
!7 
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & ?6 & 1 &!10 \\
?3 & ?4  & 4 & !8  \\
?5 & 2 & !9 \\
!7 
\end{young}}
\\[1.5ex]
\ \to \ %
{\begin{young}[c]
?1  & ?2 & ?6 & 1 &!10 \\
?3 & ?4  & ?8 & !12  \\
?5 & 2 & !9 \\
!7 
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & ?6 & 1 &!10 \\
?3 & ?4  & ?8 & !12  \\
?5 & ?9 & !13 \\
!7 
\end{young}}
\ \to \ %
{\begin{young}[c]
?1  & ?2 & ?6 & ?10 &!14 \\
?3 & ?4  & ?8 & !12  \\
?5 & ?9 & !13 \\
!7 
\end{young}}
\end{gather*}
\caption{An alternative way to compute $T_{3142}^\la+$, 
using the 
same order for the boxes of $\la-$ as the example in Figure~\ref{fig:forward}.}
\label{fig:sequence}
\end{figure}

The steps of a rectification-type algorithm can be
performed in a variety of different but equivalent orders.  
In particular, instead
of sliding the entries of $U$ through $T$, from largest to smallest,
one can reverse-slide the entries of $T$ though $U$, from smallest to
largest, (see \cite{BSS} for full details).  
Figure~\ref{fig:sequence} illustrates this in the context 
of Algorithm~\ref{alg:forward}, using the example from
Figure~\ref{fig:forward}.

This perspective not only gives a reformulation of
Algorithm~\ref{alg:forward}, but allows us generalize it 
to inputs that are not permutations.
Let $\sigma = \sigma_1\sigma_2\sigma_3 \dots$ be an infinite 
sequence, with $\sigma_k \in \{1, \dots, n\}$ for $k =1, 2, 3, \dots$.
We construct a sequence $\boxseq_\sigma^U = \one_1\, \one_2\, \one_3\, \dots$
of boxes of $\la+$, as follows.

\begin{algorithm} 
\label{alg:sequence}
\emph{INPUT:} The pair $(\sigma, U)$.
\begin{algorithmic}
\State Let $U_0 := U$;
\For {$k = 1, 2, 3, \dots$}
\State Let $U_k$ be the tableau obtained by reverse-sliding 
    box $\diag_{\sigma_k}$ through $U_{k-1}$;
\State Define $\one_k$ to be the final position of the sliding path;
\State Delete the entry in $\diag_{\sigma_k}$ from $U_k$, if one exists;
\EndFor
\State \Return $\boxseq_\sigma^U := \one_1\, \one_2\, \one_3\, \dots.$
\end{algorithmic}
\end{algorithm}

We use this sequence to
define a function $\delta_\sigma^U : \{1, \dots, n\} \to \ZZ_{\geq 0}$,
\[
  \delta_\sigma^U(i) := 
  \#\{k  \mid \sigma_k = i \text{ and }\one_k \neq \diag_i\}\,,
\]
where $\boxseq_\sigma^U = \one_1\,\one_2\,\one_3\,\dots$.  This function
will be key in proving Theorem~\ref{thm:combine}.  If
$\boxseq_\sigma^U$ is independent of $U \in \SYT(\la-)$, we write
$\boxseq_\sigma^\la+ := \boxseq_\sigma^U$, and $\delta_\sigma^\la+ := 
\delta_\sigma^U$.

Strictly speaking, Algorithm~\ref{alg:sequence} is not a proper algorithm, 
in that it does not terminate;
however, we are really only interested in a finite part
of $\boxseq_\sigma^U$.  Once $k$ is sufficiently large, we have 
$\one_k = \diag_{\sigma_k}$.  For $N \geq 0$, denote the truncation
of a sequence at its $N$\nth
term by
$\trunc_N(a_1 a_2 a_3 \dots) := (a_1 a_2 \dots a_N)$.

For $w \in S_n$,  define $w^*$ to be the repeating sequence
\[
w^* := a_1a_2 \dots a_n\, a_1 a_2 \dots a_n\, a_1 a_2 \dots,
\]
where $a_1a_2 \dots a_n$ is the word representing $w^{-1}$ in
one line notation (i.e. $a_i = w^{-1}(i)$ for $i=1, \dots, n$).  
The following proposition precisely states 
the relationship between
Algorithms~\ref{alg:forward} and~\ref{alg:sequence}.

\begin{proposition}
\label{prop:reformulate}
Write $\boxseq_{w^*}^U = \one_1\, \one_2\, \one_3\, \dots.$
For each box $\one \in \la+$, $T_w^U[\one]$ is the smallest $k$ 
such that $\one_k = \one$\,.  Hence, for some $N \geq 0$,
$\trunc_N\big(\boxseq_{w^*}^U\big)$ determines $T_w^U$.
In addition, we have 
$T_w^U[\diag_i] = w(i) + n \cdot \delta_{w^*}^U(i)$.
\end{proposition}

\begin{proof}
The first two statements are simply a more precise formulation of the
remarks at the beginning of this section.  For the third, note
that the $k$\nth term of $w^*$ is equal to $i$, for 
$k = w(i), w(i)+n, w(i)+2n, \dots.$
By the definition of $\delta_\sigma^U$, the 
$(\delta_{w^*}^U+1)$\nth term in this sequence is the smallest
$k$ such that $\one_k = \diag_i$.  The former is 
$w(i) + n\cdot \delta_{w^*}^U(i)$, and the latter is $T_w^U[\diag_i]$.
\end{proof}

Partially order the boxes of $\la+$: let $\one \leq \one'$ if $\one'$
is both weakly right of $\one$ and weakly above $\one$.  

\begin{proposition}
\label{prop:descents}
Write $\boxseq_{\sigma}^U = \one_1\, \one_2\, \one_3\, \dots.$ 
If $\sigma_k < \sigma_{k+1}$ then $\one_k < \one_{k+1}$;
if $\sigma_k > \sigma_{k+1}$ then $\one_k > \one_{k+1}$.
\end{proposition}

\begin{proof}
This follows from the fact that \emph{jeu de taquin} 
preserves horizontal
(and vertical) strips.
\end{proof}

For a sequence
$a_1 a_2 a_3 \dots $ with terms from a partially ordered set
(e.g. the numbers $\{1, \dots, n\}$ or the boxes of $\lambda_+$), 
define 
the \defn{strict Knuth transformations} to be the operations
\[
\kappa_k(a_1 a_2 a_3 \dots ) 
= a_1 a_2 \dots a_{k-1} x y z a_{k+3} a_{k+4} \dots,
\]
where
\[
    xyz = 
\begin{cases} 
a_k a_{k+2} a_{k+1}
&\quad 
\text{if\ \ $a_{k+1} < a_k < a_{k+2}$ 
\ \ or\ \ $a_{k+2} < a_k < a_{k+1}$}
\\
a_{k+1} a_k a_{k+2} 
&\quad 
\text{if\ \ $a_k < a_{k+2} < a_{k+1}$ 
\ \ or\ \  $a_{k+1} < a_{k+2} < a_k$}
\\
\text{undefined}
&\quad 
\text{otherwise.}
\end{cases}
\]
In the third case $\kappa_k(a_1a_2a_3\dots)$ is also undefined.
These are similar to elementary Knuth transformations on sequences, 
except that the inequalities are required to be strict.
We define two sequences $a_1 a_2 a_3 \dots$ and $b_1 b_2 b_3 \dots$
to be \defn{equivalent} if for every $N \geq 0$, there exists a 
finite sequence
$\kappa_{k_1}, \kappa_{k_2}, \dots, \kappa_{k_m}$ of strict 
Knuth transformations such that
\[
\trunc_N \big(\kappa_{k_1} \circ \kappa_{k_2} \circ \dots \circ \kappa_{k_m}
(a_1 a_2 a_3 \dots)\big) = \trunc_N \big(b_1 b_2 b_3 \dots\big)
\,.
\]
When $a_1a_2a_3\dots$ is a sequence of boxes, this 
generalizes of the notion of dual equivalence on tableaux \cite{Hai}.

\begin{proposition}
\label{prop:equivariant}
Let  $\sigma$ be a sequence with terms from $\{1, \dots, n\}$. 
If $\kappa_k(\sigma)$ is defined, then
$\kappa_k(\boxseq_\sigma^U)$ is defined, and
\begin{packedenumi}
\item $\boxseq_{\kappa_k(\sigma)}^U = \kappa_k(\boxseq_\sigma^U)$;
\item $\delta_{\kappa_k(\sigma)}^U(i) = \delta_\sigma^U(i)$, 
for $i =1 ,\dots, n$.
\end{packedenumi}
\end{proposition}

\begin{proof}
Write $\Hsigma := \kappa_k(\sigma)$.
Let $\boxseq_\sigma^U = \one_1\, \one_2\, \one_3\, \dots$,
and let $U_0, U_1, U_2, \dots$ be the sequence of tableaux produced
in Algorithm~\ref{alg:sequence}.
Let $\boxseq_{\Hsigma}^U 
 = \Hone_1\, \Hone_2\, \Hone_3\, \dots$,
and $\HU_0, \HU_1, \HU_2, \dots$ be the corresponding objects
for $\Hsigma$.
Since $\sigma_j = \Hsigma_j$ for $j < k$,
we have $\one_j = \Hone_j$ and $U_j = \HU_j$, for $j < k$.
In particular $U_{k-1} = \HU_{k-1}$.
Given $3$ boxes $\one, \one', \one''$ let $T(\one, \one', \one'')$
denote the standard Young tableau with entries $1$, $2$, and $3$,
in boxes $\one$, $\one'$ and $\one''$ respectively.
The pair of tableaux
$T(\diag_{\sigma_k}, \diag_{\sigma_{k+1}}, \diag_{\sigma_{k+2}})$
and
$T(\diag_{\Hsigma_k}, \diag_{\Hsigma_{k+1}}, \diag_{\Hsigma_{k+2}})$
form a dual equivalence class. %\cite[Proposition 2.2]{Hai}; 
It follows from \cite[Corollary 2.8]{Hai} that $U_{k+2} = \HU_{k+2}$,
and since $\sigma_j = \Hsigma_j$ for $j > k+2$ we have
$\one_j = \Hone_j$ for $j > k+2$.
By \cite[Lemma 2.3]{Hai}, $T(\one_k, \one_{k+1}, \one_{k+2})$
and $T(\Hone_k, \Hone_{k+1}, \Hone_{k+2})$
also form a dual equivalence class, which implies that the three-term
sequences
$(\one_k, \one_{k+1}, \one_{k+2})$ 
and $(\Hone_k, \Hone_{k+1}, \Hone_{k+2})$
are related by a strict Knuth transformation.  This proves (i),
and (ii) is straightforward.
\end{proof} 

This leads to our strategy for proving 
Theorems~\ref{thm:welldefined} and~\ref{thm:combine}, which
is outlined in the next proposition.
For the second statement in 
Theorem~\ref{thm:combine} we will need to consider 
how the constructions in Algorithms~\ref{alg:forward} 
and~\ref{alg:reverse} are related for different choices of diagonal.
This is facilitated by the following definition.
Let $\primela+/\primela-$ be another diagonal of $\Rect$\,.
Let $\one_1\,\one_2\,\one_3\,\dots$ be a sequence of boxes of $\la+$,
and let $\one'_1\,\one'_2\,\one'_3\,\dots$ be a sequence of boxes of 
$\primela+$.  We say that these sequences are \defn{compatible} if
$\one'_k = \one_k$ whenever $\one'_k \in \la+$ and $\one_k \in \primela+$.

\begin{proposition}
\label{prop:strategy}
Suppose that $\sigma$ is equivalent to $w^*$, and
$\boxseq_\sigma^\la+$ is well-defined
(i.e. $\boxseq_\sigma^U$ is independent $U \in \SYT(\la-)$).  
Then the following are true.
\begin{packedenumi}
\item  
$T_w^\la+$ is well-defined (i.e.
$T_w^U$ is independent of $U \in \SYT(\la-)$).
\item 
$T_w^\la+[\diag_i] = w(i) + n \cdot \delta_\sigma^\la+(i)$.
\item 
Suppose $\primela+$ is obtained from $\la+$ by adding one box.
If
$\boxseq_\sigma^\primela+$ is well-defined and compatible with 
$\boxseq_\sigma^\la+$, then the tableaux
$T_w^\la+$ and  $T_w^{\la+'}$ coincide on $\la+$.
\end{packedenumi}
\end{proposition}

\begin{proof}
Let $U, \HU \in \SYT(\Rect)$. To prove (i), we 
must show that $T_w^U = T_w^\HU$.
By Proposition~\ref{prop:reformulate},
there exists $N \geq 0$ so that $T_w^U$ and $T_w^\HU$ are determined by
$\trunc_N(\boxseq_{w^*}^U)$ and $\trunc_N(\boxseq_{w^*}^\HU)$; 
therefore it is enough to show that the latter two are equal.
Since $w^*$ is equivalent to $\sigma$ there is a sequence
$\kappa_{k_1}, \kappa_{k_2}, \dots, \kappa_{k_M}$ of strict 
Knuth transformations such that
\[
\trunc_N \big(\kappa_{k_1} \circ \kappa_{k_2} \circ \dots \circ \kappa_{k_M}
(\sigma)\big) = \trunc_N \big(w^*\big)
\]
Since $\boxseq_\sigma^\la+ =\boxseq_\sigma^U =\boxseq_\sigma^\HU$,
by Proposition~\ref{prop:equivariant}(i) we have
\[
\trunc_N\big(\boxseq_{w^*}^U\big)
=
\trunc_N \big(\kappa_{k_1} \circ \kappa_{k_2} \circ \dots \circ \kappa_{k_M}
(\boxseq_\sigma^\la+)\big) 
= 
\trunc_N\big(\boxseq_{w^*}^\HU\big)\,,
\]
as required.
Similarly, (ii) follows from
Proposition~\ref{prop:reformulate} and
Proposition~\ref{prop:equivariant}(ii).
For (iii), it is easy to see that $\boxseq_\sigma^{\la+}$
is compatible with $\boxseq_\sigma^{\primela+}$ if and only if 
$\kappa_k(\boxseq_{\sigma}^{\la+})$ is compatible with
$\kappa_k(\boxseq_{\sigma}^{\primela+})$.  By 
Proposition~\ref{prop:equivariant}(i), 
$\boxseq_{w^*}^\la+$ is compatible with
$\boxseq_{w^*}^\primela+$; the result follows by
Proposition~\ref{prop:reformulate}.
\end{proof}

In the next section we will construct a suitable $\sigma$ for each 
permutation $w$, enabling us to prove Theorems~\ref{thm:welldefined}
and~\ref{thm:combine}.  The construction of $\sigma$ is based on
the \emph{cyclage} operation of Lascoux and Sch\"utzenberger \cite{LS}.
The following two facts will be used to establish equivalence:

\begin{proposition}
\label{prop:periodicequivalence}
Let $w, \Hw \in S_n$ be two permutations.  
If $w^{-1}$ and $\Hw^{-1}$ have the same insertion tableau,
then $w^*$ is equivalent to $\Hw^*$.
\end{proposition}

\begin{proof}
Fix $N \geq 0$.
Let $a_1a_2 \dots a_n$ and $b_1b_2 \dots b_n$ be the words representing
$w^{-1}$ and $\Hw^{-1}$ respectively.  Since these words have the same 
insertion tableau, they are related by a finite sequence of elementary 
Knuth transformations, and since $a_i \neq a_j$ for $i \neq j$, each of
these is a strict Knuth transformation.  It follows that
$w^*$ can be transformed into any sequence of the form
\[
   b_1b_2 \dots b_n\,b_1b_2 \dots b_n\ \cdots\ b_1b_2 \dots b_n\,
   a_1a_2 \dots a_n\,a_1a_2 \dots a_n\,a_1a_2\ \dots
\]
using a finite sequence of strict Knuth transformations.
If there are at least $N/n$ copies of $b_1b_2\dots b_n$, then 
truncating
at the $N$\nth term gives $\trunc_N(\Hw^*)$, as required.
\end{proof}

\begin{proposition}
\label{prop:rowstrict}
Let $a_1a_2a_3 \dots$ and $b_1b_2b_3 \dots$ be 
positive integer sequences.
Let $A_k$ be the insertion tableau of the finite word
$a_1a_2 \dots a_k$, and let $B_k$ be the insertion tableau of
$b_1b_2 \dots b_k$.
Suppose there exists a number $M$ such that the following hold: 
$A_M = B_M$;
$a_k = b_k$ for all $k > M$; and $A_k$ and $B_k$ are row-strict 
for all
$k \leq M$.  Then $a_1a_2a_3 \dots$ is equivalent to $b_1b_2b_3 \dots.$
\end{proposition}

\begin{proof}
Let $A_{k,1}A_{k,2}\dots A_{k,k}$ denote the reading word of $A_k$.
For $k \leq M$ there is a finite sequence of
elementary Knuth transformations taking
\[
   A_{k-1,1}A_{k-1,2} \dots A_{k-1,k-1}a_k \ \mapsto\ %
   A_{k,1}A_{k,2} \dots A_{k,k-1}A_{k,k}\,;
\]
the precise sequence can be found in many references
(e.g. \cite[Section 2.1]{Ful} or \cite[Section 6.1]{LLT}).
It is easy to verify that
if $A_k$ is row-strict, then all of the transformations in this 
sequence are \emph{strict} Knuth transformations.
This shows that $a_1a_2a_3 \dots$ is equivalent to 
\[
  A_{M,1}A_{M,2}\dots A_{M,M}a_{m+1}a_{m+2} \dots
  =
  B_{M,1}B_{M,2}\dots B_{M,M}b_{m+1}b_{m+2} \dots
\]
which, by the same argument, is equivalent to $b_1b_2b_3 \dots$.
\end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\section{Descent sequences}
\label{sec:descents}

Recall that $i \in \{1, \dots, n-1\}$ is a \defn{descent} of
$w$ if $w(i) > w(i+1)$;  if $w(i) < w(i+1)$, then $i$ is 
an \defn{ascent} of $w$.
Let $\identity \in S_n$ denote the identity element, and let
$w_0 \in S_n$ denote the long element, 
$w_0 = n\,n{-1}\,\dots\,2\,1$.
The \defn{major index} of $w$ is defined
to be the sum of the descents of $w$.  For example, 
$\identity$ is the unique permutation with major index $0$, 
$w_0$ is the unique permutation with major index $n(n-1)/2$.

\begin{lemma}
\label{lem:equivalent}
Let $d_1 > d_2 > \dots > d_t$ be the descents of $w$ in decreasing
order, and let $d_i = 0$ for $i > t$.  Then
$w^*$ is equivalent to the sequence
\begin{equation}
\label{eqn:descentseq}
\sigma_{d_1d_2d_3\dots} :=
(d_1{+}1, d_1{+}2, \dots , n,\,
d_2 {+}1 , d_2 {+}2, \dots ,n,\,
d_3 {+}1 , d_3 {+}2, \dots ,n,\,
\dots)\,.
\end{equation}
\end{lemma}

\begin{proof}
For a permutation $w \in S_n$ define $\epsilon_w \in \{1, \dots, n\}$ and
permutations
$\Hw, w' \in S_n$ as follows.
Let $b_1b_2 \dots b_n$ be the reading
word of the insertion tableau of $w^{-1}$.
Let $\epsilon_w := b_1$.  Let $\Hw$ and $w'$ be the permutations 
whose inverses are represented by the words 
$b_1 b_2 \dots b_n$ and $b_2 b_3 \dots b_n b_1$, 
respectively, in one line notation.
Thus 
\[
   \big(\epsilon_w, (w')^*\big)  =
    b_1 b_2 \dots b_n \, b_1 b_2 \dots b_n \, b_1 b_2 \dotsc
     = \Hw^*\,.
\]
By Proposition~\ref{prop:periodicequivalence},
$w^*$ is equivalent to $\Hw^* = \big(\epsilon_w,(w')^*\big)$.
Using this argument repeatedly, $w^*$ is equivalent
to the sequence
\[
   \big(\epsilon_w, \epsilon_{w'}, \epsilon_{w''}, \dots,
      \epsilon_{w^{(M-1)}},
    (w^{(M)})^*\big)
\]
for any $M \geq 0$.
Since Knuth transformations preserve the descents of the inverse 
of a permutation, the descents of $w$ are the same as descents of $\Hw$.

Let $M(w)$ be the
major index of $w_0 w w_0$, which is equal to 
$(n-d_1) + \dots + (n-d_t)$.
Suppose $w \neq \identity$.  Then $\epsilon_w > 1$.  
Since $\Hw(\epsilon_w) = 1$, and $w'(\epsilon_w) = n$,
$\epsilon_w{-}1$ is a descent of $w$ and an ascent of $w'$.
If $\epsilon_w < n$, then $\epsilon_w$ is an ascent of $w$ and a 
descent of $w'$.
For $i \notin \{\epsilon_w{-}1, \epsilon_w\}$,
$i$ is a descent of $w$ if and only if $i$ is a descent of $w'$.
It follows from these remarks that $M(w') = M(w) -1$.

Since $M(w^{(M(w))}) = 0$, $w^{(M(w))} =\identity$.
Thus we have shown $w^*$ is equivalent $(w^\#, \identity^*)$, where
\[
w^\# := 
   \big(\epsilon_w, \epsilon_{w'}, \epsilon_{w''}, \dots ,
      \epsilon_{w^{(M(w)-1)}}\big)\,.
\]
We now show, by induction on $M(w)$, that $(w^\#,\identity^*)$ 
is equivalent to $\sigma_{d_1d_2d_3\dots}$.
If $w = \identity$, the result is
trivial.  Suppose $M(w)>1$ and assume the result is true 
for $w'$.  Then $(w^\#, \identity^*) = (\epsilon_w , (w')^\#, \identity^*)$
is equivalent to $(\epsilon_w, \sigma_{d'_1d'_2d'_3\dots})$,
where
$d'_1 > \dots > d'_{t'}$ are the descents of $w'$, 
and $d'_i = 0$ for $i > t'$.
The arguments above show that $\epsilon_w = d_s{+}1$ for some $s \leq t$;
if $\epsilon_w < n$, then $d'_i = d_i$ for $i \neq s$, 
and $d'_s = \epsilon_w$;
if $\epsilon_w = n$, then $d'_i = d_{i+1}$ for all $i$.
In either case, $\sigma_{d'_1d'_2d'_3\dots}$ is 
obtained from $\sigma_{d_1d_2d_3 \dots}$ by deleting the 
first occurrence of $\epsilon_w$.
With this in mind, it follows readily from 
Proposition~\ref{prop:rowstrict}
that $(\epsilon_w, \sigma_{d'_1d'_2d'_3\dots})$  
is equivalent to $\sigma_{d_1d_2d_3\dots}$.
\end{proof}

\begin{lemma}
\label{lem:welldefined}
Assume that $n$ is the number of columns of\/ $\Rect$\,.
Let $\sigma = \sigma_{d_1d_2d_3 \dots}$ 
be the sequence in~\eqref{eqn:descentseq}. 
\begin{packedenumi}
\item 
$\boxseq_\sigma^\la+$ is well-defined:  the $k$\nth box of this 
sequence is in column $c_k$, where
\[
c_1 c_2 c_3 \dotsc  := 
(1, 2, \dots, n{-}d_1, \,
1, 2, \dots, n{-}d_2, \,
1, 2, \dots, n{-}d_3, \,
\dots)\,.
\]
\item
If $C_i$ is the length of the $i$\nth column of 
$\la+$, then
\[
\delta_\sigma^\la+(i) = C_i
- \#\big\{j \bigmid d_j \geq i\big\} 
+ \#\big\{j \bigmid d_j \geq n{+}1{-}i\big\} 
- 1
\qquad\text{for $i = 1, \dots, n$.}
\]
\item
For any other diagonal $\primela+/\primela-$ of\/ $\Rect$\,, 
$\boxseq_\sigma^\primela+$ is compatible with $\boxseq_\sigma^\la+$.
\end{packedenumi}
\end{lemma}

\begin{proof}
We introduce the notation 
\[[i,j] := i + \sum_{s=1}^{j-1} (n-d_s)\,,
\qquad\text{for $1 \leq i \leq n{-}d_j$ and $j \geq 1$}. 
\]
Thus $c_{[i,j]} = i$ and $\sigma_{[i,j]} = d_j+ i$ for all $i, j$.
Fix $U \in \SYT(\la-)$, 
and write $\boxseq_\sigma^U = \one_1\ \one_2\ \one_3 \dots.$
Suppose $\one_k$ is in column $e_k$.
To prove (i), we need to show that $e_{[i,j]} = i$
for all $i,j$.  
%(If each box is in the correct column, then
%it is the correct sequence.)
We will do this by strong induction on $j$.  

Fix $j \geq 0$. 
Assume that $e_{[i,s]} = i$ for 
$1 \leq s \leq j$, $1 \leq i \leq n-d_s$.
Let $d_0 := n$, $j' := j+1$, $p := n - d_j$, $p' := n-d_{j'}$.
We will prove the following: 
\begin{packedenum}
\item[(a)]
$e_{[i,j']} \geq i$ for $1 \leq i \leq p'$; 
\item[(b)]
$e_{[i,j']} = e_{[i-1,j']}+1$ for $p < i \leq p'$
(where $e_{[0,1]} := 0$);
\item[(c)]
$e_{[i,j']} \leq e_{[i,j]}$ for $1 \leq i \leq p$.
\end{packedenum}
These imply that $e_{[i,j']} = i$ for $i = 1,\dots, p'$.

Since 
\[\sigma_{[1,j']}\sigma_{[2,j']}\dots\sigma_{[p',j']}
 = (d_{j'}{+}1 < d_{j'}{+}2 < \ldots < n)\,,
\]
by Proposition~\ref{prop:descents}
we have 
$\one_{[1,j']} < \one_{[2,j']} < \dots < \one_{[p',j']}$.
Because of the order in which the slides are performed,
$\one_{[i+1,j']}$ cannot be above $\one_{[i,j]}$ and in the same column.
It follows that 
$e_{[1,j']} < e_{[2,j']} < \dots < e_{[p',j']}$,
which proves (a).
%;
%hence $e_{[i,j']} \geq i$, for $i = 1, \dots, p'$.
Since all boxes $\one_1\,, \dots,\,\one_{[p,j]}$ are
in the first $p$ columns, any box $\one_{[i,j']}$ which is not in
the first $p$ columns must be in the first row.  In particular
this applies when $i > p$, which proves (b).
%$\one_{[p+1,j']}\,, \dots,\, \one_{[p',j']}$
%must be in the first row,
%which shows that $e_{[i,j']} = e_{[i-1,j']}+1$ for $i > p$.

If $p' = n$, then (c) follows immediately from (a).  To complete
the proof of (i), suppose that $p' < n$.  Then $d_j > d_{j'}$.
Let $a_i := \sigma_{[i,j]} = d_j+i$ 
and let $b_i := \sigma_{[i,j']} = d_{j'}+i$.
Consider sequence obtained from $\sigma$ by changing the subsequence
of length $2p$ starting at $\sigma_{[1,j]}$ from
\[
(a_1< a_2 < \ldots < a_p > b_1< b_2 < \dots < b_p)
\]
to
\[
   (a_1 > b_1 < a_2 > b_2 < \ldots < a_p > b_p)\,.
\]
This transformation can be realized as
$K_{p-1} \circ K_{p-2} \circ \dots \circ K_1(\sigma)$,
where
\[
K_i := \kappa_{[1,j]+2i-2} \circ \kappa_{[1,j]+2i-1} \circ \kappa_{[1,j]+2i} \dots 
\circ \kappa_{[1,j]+i+p-3}
\]
is the composition of strict Knuth transformations that
moves $b_i$ next to $a_i$.  
For example, $K_1$ performs the following sequence of transformations:
\begin{align*}
(a_1 <  &\ldots< a_{p-2} < a_{p-1}< a_p > \boldsymbol{b_1}
< b_2 < \ldots < b_p)
\\
& 
\mapsto \ %
(a_1 <  \ldots < a_{p-2} < a_{p-1} > \boldsymbol{b_1} < a_p
> b_2 < \ldots < b_p)
\\
&
\mapsto \ %
(a_1 <  \ldots < a_{p-2} > \boldsymbol{b_1} < a_{p-1} < a_p
> b_2 < \ldots < b_p)
\\
&\dots\\
& 
\mapsto \ %
(a_1 > \boldsymbol{b_1} < a_2 < \ldots < a_p > b_2 < \ldots < b_p)\,.
\end{align*}
Here we have recorded only the subsequence of length $2p$ starting
at $[1,j]$ --- the remaining terms are unaffected by these transformations.

Let $\alpha_i := \one_{[i,j]}$ and let $\beta_j := \one_{[i,j']}$.
The corresponding subsequence of $\boxseq_\sigma^U$ is
\[
   (\alpha_1 < \alpha_2 < \ldots < \alpha_p > \beta_1 < \beta_2 < \ldots < \beta_p)\,.
\]
By Proposition~\ref{prop:equivariant}, 
$K_{p-1} \circ K_{p-2} \circ \dots \circ K_1(\boxseq_\sigma^U)$ is defined,
and by Proposition~\ref{prop:descents}, each strict Knuth 
transformation must produce a
sequence with the correct descent pattern.
Using these two facts, one can deduce (by a straightforward
inductive argument) 
that for all $r  = 1, \dots, p-1$,
the corresponding subsequence of
$K_r \circ \dots \circ K_1(\boxseq^U_\sigma)$ must be of the form
\[
(\alpha_{q_1} > \gamma_1 < \alpha_{q_2} < \gamma_2 > 
\ldots < \alpha_{q_r} > \gamma_r < \gamma_{r+1} < \ldots < \gamma_p 
> \beta_{r+1} < \ldots < \beta_p)\,,
\]
where $1 \leq q_1 < q_2 < \dots < q_r \leq p$ 
and $(\gamma_1 < \gamma_2  < \cdots < \gamma_p)$ is obtained from
$\alpha_1\alpha_2 \dots \alpha_p$ by replacing $\alpha_{q_i}$ 
replaced by $\beta_i$ for
$i =1, \dots,r$.   In particular, when $r = p-1$, we have $\gamma_p > \beta_p$;
thus $\gamma_p = \alpha_p$, and $q_i = i$ for all $i$.
This shows that
\[
K_{p-1} \circ \dots \circ K_1(\boxseq_\sigma^U)
= (\dots\ \alpha_1 > \beta_1 < \alpha_2 > \beta_2 < \ldots < \alpha_p > \beta_p \ \dots)\,.
\]
The descent pattern of this sequence establishes that $\alpha_i > \beta_i$, 
which proves (c).
%which shows that $e_{[i,j']} \leq e_{[i,j]}$,
%for all $i =1, \dots, p$. 

For (ii), suppose that $C_i$\nth occurrence of $i$ in
the sequence $c_1c_2c_3 \dots$ occurs at $[s,i]$.
Since the subsequence 
\[
c_{[1,j]} c_{[2,j]}\dots c_{[n-d_j,j]} = (1,2, \dots, n{-}d_j)
\]
excludes $i$ if and only if $d_j \geq n+1-i$, 
$s = C_i + \#\{j \mid d_j \geq n{+}1{-}i\}$.
By (i), $\one_k$ is in column $c_k$, if and only if $k=[j,i]$
for some $j$, and $\one_k = \diag_i$ when $j \geq s$;
therefore
$\delta_\sigma^{\la+}(i)$ is the number of occurrences of $i$ 
in the sequence $\trunc_{[n-d_{s-1},s-1]}(\sigma)$.
Since the subsequence
\[
\sigma_{[1,j]} \dots \sigma_{[n-d_j,j]}
 = (d_j{+}1, d_j{+}2, \dots, n)
\]
excludes $i$ if and only if $d_j \geq i$, 
$\delta_\sigma^\la+(i) = s-1 - \#\{j \mid d_j \geq i\}$, 
as required.

Finally, (iii) follows immediately from (i).
\end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Proofs}
\label{sec:proofs}
We now prove Theorems~\ref{thm:welldefined}, \ref{thm:combine} 
and \ref{thm:bijection}.

\begin{proof}[Proof of Theorem~\ref{thm:welldefined}]
Since the result is symmetrical with respect to rows and columns,
we may assume, without loss of generality, that $n$ is the number 
of columns of $\Rect$\,.
By Lemma~\ref{lem:equivalent} and Lemma~\ref{lem:welldefined}(i),
$w^*$ is equivalent to a sequence $\sigma$ such that $\boxseq_\sigma^\la+$
is well-defined.  The theorem 
therefore follows from Proposition~\ref{prop:strategy}(i).
\end{proof}

\begin{proof}[Proof of Theorem~\ref{thm:combine}]
Again, assume, without loss of generality, that $n$ is the number 
of columns of $\Rect$\,.  
Using Lemma~\ref{lem:equivalent}, Lemma \ref{lem:welldefined}(ii), 
Proposition~\ref{prop:strategy}(ii), and Proposition~\ref{prop:reformulate},
we compute that
\[
  T_w^\la+[\diag_i] 
   = w(i) + n\cdot\Big(
   C_i -
   \#\{j \mid d_j \geq i\}
   + \#\{j \mid d_j \geq n{+}1{-}i\} - 1 \Big)\,,
\]
where $d_1 > d_2 > \dots > d_t$ are the descents of $w$, and
 $C_i$ is the length of column $i$ in $\la+$.
For a partition $\lambda \subset \Rect$\,,
with row lengths $(\lambda_1, \dots, \lambda_m)$,
let $\lambda^\vee$ denote the partition  with row lengths
$(n-\lambda_m, \dots, n-\lambda_1)$.
If $\one$ is the
box of $\Rect$ in column $i$ and row $j$, let $\one^\vee$ denote the
box in column $n+1-i$ and row $m+1-j$.   For a skew tableau $T$
of shape $\lambda/\mu \subset \Rect$\,, let $T^\vee$ denote the 
tableau of shape $\mu^\vee/\lambda^\vee$ with entries 
$T^\vee[\one] := mn+1-T[\one^\vee]$.
The relationship between Algorithms~\ref{alg:forward}
and~\ref{alg:reverse} is
\begin{equation}
\label{eqn:forwardreverse}
   T_w^{\Rect/\la-} = (T_{w_0ww_0}^\veela-)^\vee\,.
\end{equation}
Note that
$m+1 - C_{i}$ is the length of column $n+1-i$ in $\veela-$,
and 
$n{-}d_1 < n{-}d_2 < \dots < n{-}d_t$ are the descents of $w_0ww_0$.
We compute:
\begingroup
\allowdisplaybreaks[4]
\begin{align*}
T_w^{\Rect/\la-}[\diag_i] 
&= (T_{w_0ww_0}^\veela-)^\vee[\diag_i]  \\
&= mn+1 - T_{w_0ww_0}^\veela-[\diag_i^\vee] 
\\
&= mn+1 - \Big(w_0ww_0(n{+}1{-}i) 
+ n \cdot \delta_{(w_0ww_0)^*}^\veela-(n{+}1{-}i)\Big) \\
&= 
mn+1 - \left(
n{+}1{-}w(i) 
+ n \cdot \left(
\begin{aligned}
(m{+}1{-}C_i)  
%+ \#\{j \mid 1 \leq n{-}d_j \leq n{+}1{-}i\}&
+ \#\{j \mid n{-}d_j \geq n{+}1{-}i\}&
\\
%-\ \#\{j \mid 1 \leq n{-}d_j \leq i\} &
-\ \#\{j \mid n{-}d_j \geq i\} &
- 1
\end{aligned}
\right)
\right)
\\
&= w(i) + n\cdot\Big(
   C_i -
   \#\{j \mid n{-}d_j \geq n{+}1{-}i\}
   + \#\{j \mid n{-}d_j \geq i\} - 1 \Big)
\\
&=
T_w^\la+[\diag_i]\,,
\end{align*}
\endgroup
i.e. $T_w^\la+$ and $T_w^{\Rect/\la-}$ agree on $\la+/\la-$.
For each vertically adjacent pair of boxes, either both are in $\la+$
or both are in $\Rect/\la-$.  Thus the agreement on $\la+/\la-$
shows that $T_w$ is column strict.  It also follows now, from
Lemma~\ref{lem:equivalent},
Lemma~\ref{lem:welldefined}(iii) and Proposition~\ref{prop:strategy}(iii),
that $T_w$ is independent of the choice of $\la+$.
Thus we may we assume $\la+ = (n, n{-}1, \dots, 2,1)$, which allows
us to see that $T_w$ is row-strict.

Let $D := \{T_w[\diag_i] \mid i = 1, \dots, n\}$ be the
set of diagonal entries of $T_w$.  Algorithm~\ref{alg:forward}
ensures that $T_w[\diag_i] \equiv w(i) \pmod n$, so $D$
contains one number from each congruence class, modulo $n$.
%Furthermore we see that $T_w[\diag_i \geq 1$ 
%from Algorithm~\ref{alg:forward}, $T_w[\diag_i] \leq mn$ 
%by Algorithm~\ref{alg:reverse}.
Since the entries of $T_w^\la+$ are 
\[
\big\{k \geq 1 \bigmid k+nj \in D \text{ for some }j \geq 0\big\}
\]
and the entries of $T_w^{\Rect/\la-}$ are
\[
\big\{k \leq mn \bigmid k-nj \in D \text{ for some }j \geq 0\big\}\,,
\]
we see that every number in $\{1, 2,\dots, mn\}$ is an entry of $T_w$.
Therefore $T_w$ is a standard Young tableau.
\end{proof}


\begin{proof}[Proof of Theorem~\ref{thm:bijection}]
For (i), we may assume that $\la+ = (n, n{-}1, \dots, 2,1)$.
This ensures that the sliding path of promotion on any $T \in \SYTRect$
passes through exactly one box of $\la+/\la-$.
Suppose that $T_w^\la+$ is computed using Algorithm~\ref{alg:forward}
by a sequence of slides whose
first step moves the entry $1$.  After this first step, if we delete 
the entry $1$ and decrement all entries by $1$, we are computing 
$T_{wc}^\la+$ instead.  This can be done immediately, or at any
point during  Algorithm~\ref{alg:forward}.
Compare this with the behaviour of  $\promote : T_w \to \promote T_w$ 
on the entries in $\la+$.  Suppose the sliding path passes through
$\la+/\la-$ at $\diag_s$.  
Since the first two steps of promotion delete 
the entry $1$ and decrement all entries by $1$, this produces
the penultimate step in the construction of $T^{\la+}_{wc}$.  Next, we
slide the empty box in the upper-left corner of $\Rect$ through the
tableau, which is the almost the same as the final of step 
in the construction of $T^{\la+}_{wc}$, except that
do not yet know what number will appear in $\promote T_w[\diag_s]$.
This shows that for all boxes of $\la+$, with
the possible exception of $\diag_s$, $T_{wc}$ coincides with 
$\promote T_w$.  Note that $s$ is the unique number such that
$T_{wc}[\diag_s] - T_{w}[\diag_s] \neq -1$.

Applying the same argument to $\Rect/\la-$ and the sliding path of
$\promote^{-1} : T_{wc} \to \promote^{-1} T_{wc}$, we see that
with the possible exception of one box $\diag_{s'}$,
$T_w$ coincides with 
$\promote^{-1} T_{wc}$ on $\Rect/\la-$.
Since $s'$ is the unique number such that 
$T_{wc}[\diag_{s'}] - T_{w}[\diag_{s'}] \neq -1$, we must have
$s = s'$.  This shows that these two sliding paths
are in fact inverse to each other, and hence $\promote T_w = T_{wc}$.

Since $T_w[\diag_i] \equiv w(i) \pmod n$,
(ii) is immediate.

To prove (iii), we use another reformulation
of Algorithm~\ref{alg:forward}.

\begin{algorithm}
\label{alg:likepromotion} \emph{INPUT:} A permutation $w \in S_n$. 
\begin{algorithmic}
\State
Begin with $T := \varnothing$, the empty tableau, and $\mu := \Rect$\,;
\While{$\mu$ is not the empty partition}
\State  Choose a corner box $\one \in \mu$;
\If {$\one = \diag_i$ for some $i$} 
   \State Set $T[\diag_i] := w(i)$;
\EndIf
\If {$\one \in \la-$}
\State Let $T'$ be the tableau obtained by sliding $\one$ through $T$; 
\State If the final position of the sliding path is $\diag_i$, then set 
$T'[\diag_i] := T[\diag_i] + n$; 
\State  Set $T := T'$;
\EndIf
\State Delete the box $\one$ from $\mu$;
\EndWhile
\State \Return the resulting tableau, $T_w^\la+ := T$.
\end{algorithmic}
\end{algorithm}

It is clear that Algorithm~\ref{alg:likepromotion} is equivalent to
Algorithm~\ref{alg:forward}: when $\one \notin \la+$ nothing happens;
when $\one \in \la+/\la-$ we create the initial entries of $T$;
when $\one \in \la-$ we proceed exactly as before.
%The key observation is that promotion behaves
%much like Algorithm~\ref{alg:likepromotion}.

Suppose $T \in \calO_n$.  For $i,k = 1, \dots, n$, let
$\Delta_{ik} := \promote^kT [\diag_i] - \promote^{k-1}T[\diag_i]$.
Thus $\Delta_{ik} \geq 0$ if and only if
the sliding path of $\promote : \promote^{k-1} T \mapsto \promote^k T$,
passes through $\diag_i$, and $\Delta_{ik} = -1$ otherwise.
The former can happen for at most one value of $i$.
Since $\promote^n T = T$, 
\[
   \Delta_{i1} + \Delta_{i2} + \dots \Delta_{in} = 0\,.
\]
Therefore, for each $i$, 
there must be at least one $k$ such that $\Delta_{ik} \geq 0$.
It follows that for each $k$ there is exactly one $i$ such that
$\Delta_{ik} \geq 0$, and for each $i$ there is exactly one $k$ such that
$\Delta_{ik} \geq 0$.  From this we see that 
if $\Delta_{ik} \geq 0$ then $\Delta_{ik} = n-1$, and therefore,
for all $k \geq 0$, 
\[
   \promote^kT [\diag_i] - (mn-k) \equiv w(i)  \pmod n\,.
\]

For $k=1, \dots, mn$, construct a tableau $T_k$ by starting with
$\promote^k T$,
subtracting $mn-k$ from all entries, and deleting any entries for which
the result is
less than or equal to $0$.  Let $\mu_k$ be the shape formed by the
unfilled boxes of $T_k$.  Thus $T_0$ is empty, $\mu_0 = \Rect$\,, and 
$T_k$ is obtained by $T_{k-1}$ as follows:
let $\one_k \in \mu_{k-1}$ be the corner of $\mu_{k-1}$ on the sliding
path of 
$\promote : \promote^{k-1}T \mapsto \promote^k T$;
slide $\one_k$ through $T_{k-1}$; add entry $k$ in the lower-right corner
of $\Rect$.

Let $w: \{1, \dots, n\} \to \{1, \dots, n\}$ be the function defined 
in the statement of (iii).
If $T_k[\diag_i]$ is non-empty, then $T_k[\diag_i] \equiv w(i) \pmod n$.
Since $\Delta_{ik} < n$ for all $k$, if 
$\one_k = \diag_i$ then $T_k[\diag_i] \leq n$, i.e. $T_k[\diag_i] = w(i)$;
and if $\one_k \in \la-$ and the sliding path of $\one_k$ passes 
through $\diag_i$, then $T_k[\diag_i] = T_{k-1}[\diag_i] + n$.
Thus if we restrict the sequence $T_0, T_1, \dots, T_{mn}$ to $\la+$,
we obtain precisely a sequence of tableaux produced by 
Algorithm~\ref{alg:likepromotion}.  Since $\promote^{mn}T = T$,
this shows that $T_w^\la+$ is the restriction of $T$ to $\la+$.
Since $T$ has no repeated entries, $w \in S_n$.
By a similar argument $T_w^{\Rect/\la-}$ is the restriction of $T$ to 
$\Rect/\la-$.  Thus $T = T_w$.  
\end{proof}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Remarks}
\label{sec:remarks}

Here is another way to compute $T_w^\la+$.  Assume that
%an $n \times m$ rectangle (i.e. 
$n$ is the number of rows of $\Rect$\,.
Define the \defn{augmented word} of $w$ to be:
\begin{align*}
\aug(w)\, := \,
  &w(1), w(1)+n, w(1)+2n, \dots w(1)+(m-1)n, \\
  &w(2), w(2)+n, w(2)+2n, \dots w(2)+(m-1)n, \\
  & \; \dots \\
  &w(n), w(n)+n, w(n)+2n, \dots w(n)+(m-1)n\,. \\
\end{align*}


\begin{theorem}
\label{thm:insertion}
$T_w^\la+$ is the insertion tableau of $\aug(w)$
restricted to $\la+$.
\end{theorem}

\begin{proof}
Let $\primela+ := (m,m+1,m+2, \dots, m+n-1)$, and
compute $T_w^{\primela+}$ using Algorithm~\ref{alg:forward}.
Choose a sequence of boxes beginning with $m-1$ boxes from
row $n$, followed $m-1$ boxes from row $n-1$, and so on.
%$\dots$ , followed 
%by $m-1$ boxes from row $1$.
(After the $m-1$ boxes from row $1$, the last $n(n-1)/2$ boxes may 
be taken in any order.)
The first $(m-1)n$ slides produce a tableau whose reading word is
$\aug(w)$.  Therefore if we restrict $T_w^{\primela+}$ to entries
$1, 2, \dots, mn$, we obtain the insertion tableau of $\aug(w)$.
By Theorem~\ref{thm:combine}, $T_w^\la+$ can be obtained as
the restriction of $T_w^\primela+$ to $\la+$. Since the entries of
$T_w^\la+$ are a subset of $\{1, 2, \dots, mn\}$, the result follows.
\end{proof}

Theorem~\ref{thm:insertion} provides an alternate definition 
of $T_w^\la+$. It has the advantage of being well-defined, and 
%defined in terms of a well-studied construction.
%As an example of how this might help, we note that one 
Theorem~\ref{thm:combine} can be proved by using Greene's 
theorem \cite{Gre} to compute the entries $T[\diag_i]$,
%and thereby deduce Theorem~\ref{thm:combine} 
(see \cite[Section 5.2]{Rhe}). 
Unfortunately, things start to break down at the proof of
Theorem~\ref{thm:bijection}, which is
intimately connected to Algorithm~\ref{alg:forward}.
The problem is that although the first and last steps of 
Algorithm~\ref{alg:forward} are related to $\aug(w)$,
the intermediate steps may not be.  
For instance, if $T$ is a tableau from one of the intermediate steps 
it is tempting to define $\aug(T)$ to be the tableau obtained by 
adding entries
$T[\diag_i]+n, T[\diag_i]+2n, \dots w(i)+(m-1)n$.
to the right of $\diag_i$.  Unfortunately, it is \emph{not true}
the Knuth class of $\aug(T)$ is invariant for all $T$.  
There are a number of variations on this idea, and none of them
appear to work.
We do not know how to construct an invariant of
Algorithm~\ref{alg:forward}, analogous to the Knuth class of
the reading word.
In particular, the intermediate tableaux in Algorithm~\ref{alg:forward} 
are not produced by ordinary \emph{jeu de taquin} 
in any obvious way.
This makes it difficult to prove Theorem~\ref{thm:bijection}, if one 
takes Theorem~\ref{thm:insertion} as the definition of $T_w$.

Another way in which our situation behaves quite differently from
ordinary rectification concerns dual equivalence.
Consider a generalization of Algorithm~\ref{alg:sequence}, in which
we allow $U \in \SYT(\la-/\mu)$ to be a skew shape, but otherwise
the algorithm is performed the same way.
This generalization \emph{does not} have the property that
$\boxseq_{w^*}^U = \boxseq_{w^*}^\HU$, when $U$ is dual equivalent
to $\HU$.  If this were true, it would provide a more straightforward 
proof of
Theorem~\ref{thm:welldefined}.  We do not know a set of elementary 
relations that generate the equivalence relation $U \sim \HU$ $\iff$ 
$\boxseq_{w^*}^U = \boxseq_{w^*}^\HU$ for all $w \in S_n$. 

Despite the aforementioned difficulties, 
Theorem~\ref{thm:insertion} can be used as a definition of
$T_w^\la+$ when $\la+$ is an \emph{arbitrary} partition
with at most $n$ rows --- even in cases where Algorithm~\ref{alg:forward}
does not make sense.
In particular, we can sometimes use this idea to define 
$T_w$, when $m < n$.
We illustrate this with an example.  
Take $n=3$, $m=2$, $\la+ = 211$,
$\la- = 1$, and $w = 132$.  The insertion tableau of $\aug(w)$ is
\begin{align*}
{\begin{young}[c]
1 & 2 & 5 \\
3 & 6 \\
4 
\end{young}}
&\qquad\Longrightarrow\qquad
%\[
T_w^\la+ = 
{\begin{young}[c]
1 & 2  \\
3 \\
4 
\end{young}}\ \ .
\\
\intertext{Similarly, using~\eqref{eqn:forwardreverse} as the definition, 
we compute:}
T_w^{\Rect/\la-} = 
{\begin{young}[c]
, & 2  \\
3 & 5 \\
4 & 6
\end{young}}
&\qquad\Longrightarrow\qquad
T_w =
{\begin{young}[c]
1 & 2  \\
3 & 5 \\
4 & 6
\end{young}}\ \ ,
\end{align*}
%
%Take $n=4$, $m=3$, $\la+ = 3211$,
%$\la- = 21$, and $w = 3142$.  The insertion tableau of $\aug(w)$ is
%\[
%\begin{young}
%1 & 2 & 6 & 10\\
%3 & 4 & 8 & 12  \\
%5 & 9 \\
%7 & 11
%\end{young}
%\]
%which we restrict to the shape $\la+$ to obtain:
%\[
%T^\la+_w = 
%\begin{young}[c]
%1 & 2 & 6 \\
%3 & 4  \\
%5   \\
%7
%\end{young}\ \ .
%\]
%Similarly, using~\eqref{eqn:forwardreverse} as a definition, 
%we can compute:
%\[
%T_w^{\Rect/\la-} = 
%\begin{young}[c]
%, & , & 6\\
%, & 4 & 8 \\
%5 & 9 & 10 \\
%7 & 11 & 12
%\end{young}\ \ .
%\]
%As these tableaux agree on $\la+/\la-$, it is reasonable to define
%\[
%T_w = 
%\begin{young}[c]
%1 & 2 & 6\\
%3 & 4 & 8 \\
%5 & 9 & 10 \\
%7 & 11 & 12
%\end{young} 
%\]
which, indeed, has order $3$ under promotion.
This is very suggestive, 
but it is unclear what to do with the proof 
of Theorem~\ref{thm:bijection}, when $m<n$.  

%In general, one could hope to prove Rhoades' cyclic sieving theorem
%combinatorially as follows.  It is known that the set $\calR_s$
%of $s$-ribbon tableaux of shape $\Rect$ (also called $s$-rim-hook
%tableaux) has cardinality $F(e^{2\pi i/s})$, where $F(q)$ is
%as in Section~\ref{sec:intro} \comment{reference}.  
%Thus the theorem could be proved by finding a bijection between
%$\calO_r$ and $\calR_s$, where $rs = mn$.  In the case where $r=n$,
%$s=m$, there is a simple bijection between $\calR_m$ and $S_n$ 
%\comment{reference}, and hence Theorem~\ref{thm:bijection} provides 
%one case of this.  More generally, $\calR_s$ can be identified with a 
%set of standard Young tableaux of skew shape.

In the thesis \cite{Rhe}, the second author observed that
a procedure based on rectification can be used construct the
set $\calO_{mn/2}$, when one of $m$, $n$ is even.
%such a bijection
%when $s=2$, and $n$ is even.  
%Let $m_1 = \lceil m/2 \rceil$,
%$m_2 = \lfloor m/2 \rfloor$, $n_1 = n/2$.  For partitions
%$\mu = (\mu_1, \dots, \mu_{m_1})$ and $\nu = (\nu_1, \dots, \nu_{m_2})$,
%with at most $m_1$ and $m_2$ parts respectively,
%left 
%$\mu \# \nu := (\mu_1+m_2, \dots, \mu_{m_1}+m_2, \nu_1, \dots, \nu_{m_2})$.
%Given $S \in \SYT(n_1^{m_1} \# n_1^{m_2} / n_1^{m_1})$ construct
%a tableau $T_S$ as follows: rectify $S$ and reverse rectify $S$;
%add $mn/2$ to each entry of the latter and superimpose these 
%two tableaux.
%For example,
%\begin{gather*}
% S =
%{\begin{young}[c]
%, & , & 4 & 6 \\
%, & , & 5 & 8 \\
%1 & 2 \\
%3 & 7 \\
%\end{young}}
%\qquad
%\Longrightarrow
%\qquad
%\Rect(S) = 
%{\begin{young}[c]
%1 & 2 & 4 & 6 \\
%3 & 5 & 8  \\
%7  \\
%,
%\end{young}}
%\qquad
%\Rect^*(S) = 
%{\begin{young}[c]
%, & , & , & , \\
%, & , & , & 4 \\
%, & 2 & 5 & 6 \\
%1 & 3 & 7 & 8
%\end{young}}
%\\
%T_S = 
%{\begin{young}[c]
%1 & 2 & 4 & 6 \\
%3 & 5 & 8 & 12 \\
%7 & 10& 13 & 14 \\
%9 & 11 & 15 & 16
%\end{young}}
%\end{gather*}
%
%\begin{theorem}
%$S$ \mapsto $T_S$ is a bijection between $\calR_2$ and $\calO_{mn/2}$.
%\end{proof}
%
In this case, other bijections are known,
%for describing the elements of 
%$\calO_{mn/2}$ 
(see \cite[Proposition 3.10]{Pur-ribbon}); 
however it is not obvious that they are equivalent.  
This provides a new perspective, and gives further hints that the methods
introduced in this paper may apply beyond the case of minimal 
orbits.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\bigskip

%\footnotesize%
%   %\textit{Address}, K.~Purbhoo:
%   \textsc{K. Purbhoo, Combinatorics and Optimization Department, 
%       University of Waterloo, 200 University Ave. W.,  Waterloo, 
%       ON, N2L 3G1, Canada.} \texttt{kpurbhoo@uwaterloo.ca}. 
%
%   \medskip
%
%   %\textit{Address}, D.~Rhee:
%   \textsc{Operations Research Center, MIT, 77 Mass Ave., Building E40--130,
%     Cambridge, MA, 02139-4307, USA.}
%\texttt{donguk@mit.edu}.
%\textsc{D. Rhee, Lunarch Studios, 144 Park Street, Unit 1702,
%Waterloo, ON, N2L 0B6, Canada.}
%\texttt{david@lunarchstudios.com}.

\end{document}

%%%%%%%%%%%%%%%%

